Author's response to reviews

Title: Quality of life of Lithuanian women with early stage breast cancer

Authors:

Giedre Bulotiene (giebul@yahoo.com)
Jonas Veseliunas (jonas.veseliunas@loc.lt)
Valerijus Ostapenko (kruties.chirurgija@loc.lt)

Version: 4 Date: 21 April 2007

Author's response to reviews: see over
21 April 2007
Dr Lolu da-Silva
Assistant Editor
BMC-series Journals
BioMed Central
Re: MS: 1995616612104954 (BioMed Central Public Health)
Quality of life of Lithuanian women with early stage breast cancer.

Dear Dr. Lolu da Silva:

My co-authors and I were pleased to receive the reviews of our manuscript. We appreciate the thorough review that our manuscript has received.

We have followed your recommendation and have added a “Competing interests” section (page 11): “The authors declare that they have no competing interests”.

Please consider the attached revision, which incorporates the following changes in response to the reviews by Mr. Santosh K Chaturvedi and Mr. Kevin Fontaine:
Author's response to reviews
Title: Quality of life of Lithuanian women with early stage breast cancer
Authors:
Giedre Bulotiene – giebul@yahoo.com;
Jonas Veseliunas – jonas.veseliunas@loc.lt;
Valerijus Ostapenko – kruties.chirurgija@loc.lt
Version: 4 Date: 21 April 2007

Reviewer 1 (Santosh K Chaturvedi)
Discretionary Revisions

Santosh K Chaturvedi: There is need to check the quality of written English, which can be done by the authors or the editorial assistants.

Response: We have made some revisions with our English, correcting style and language without changing the content of the text.

Reviewer 2 (Kevin Fontaine)

Reviewer had no general, major, minor or discretionary remarks and was satisfied with the revision.

We thank the reviewers for their diligence.

Sincerely,

Giedre Bulotiene, MD
Institute of Oncology,
Vilnius University
Lithuania